Author's response to reviews

Title: Development of a theory-based, peer support intervention to promote weight loss among Latina immigrants

Authors:

Andrea L Cherrington (cherrington@uab.edu)
Amanda Willig (mandyrd@uab.edu)
April A Agne (aagne@uab.edu)
M. Cecilia Fowler (cescll@live.com)
Gareth Dutton (gdutton@uab.edu)
Isabel Scarinci (scarinci@uab.edu)

Version: 5  Date: 22 January 2015

Author's response to reviews:

1. We notice that you are reporting a clinical trial but have not cited a trial registration number. This must be obtained before we can begin peer review of your manuscript.

   Response: This study is now registered with clinicaltrials.gov, Trial registration: Clinical Trials NCT023444212. Registered 21 January 2015.

2. Research involving human subjects (including human material or human data) that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee.

   Response: An Ethics section was added to methods to address this issue.